Author's response to reviews

Title: Abnormal energy regulation in early life: Childhood gene expression may predict subsequent chronic mountain sickness

Authors:

Luis Huicho (lhuicho@gmail.com)
Guoqiang Xing (gxing99@yahoo.com)
Clifford Qualls (clifford.qualls@gmail.com)
Maria Rivera-Ch (mrivera@upch.edu.pe)
Jorge L Gamboa (jlgamb3@email.uky.edu)
Ajay Verma (ajay_verma@merck.com)
Otto Appenzeller (OttoArun12@aol.com)

Version: 6 Date: 24 October 2008

Author’s response to reviews:

Lima, 24 October, 2008

REF: MS: 1503895877211738

Dear Editors,

Thanks for your editorial support. I am submitting now the final version of the manuscript. All revisions were made to the most recent version you sent to me.

The changes are:

1. Tables are cited now consecutively in the text.
2. On page 10, last paragraph, first line, “PDP4” was erroneously written and thus it was replaced by “PDK4”.
3. Table 4 was missing in the former revised manuscript, and thus I have included it now in the final version.
4. I checked the references. They are listed now in the order that they follow in the manuscript and they conform to BMC Pediatrics format.

Yours sincerely,

Luis Huicho